

Geophysical Research Letters^{*}

RESEARCH LETTER

10.1029/2022GL100096

Key Points:

- The temperature sensitivity of
 photosynthesis significantly declined
 during the last two decades
- The nonlinear thermal responses of photosynthesis was the main driver of the attenuated positive effect of warming
- The temperature-limited northern ecosystems may be approaching the temperature tipping point

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

G. Yin, yingf@swjtu.edu.cn

Citation:

Yin, G., Verger, A., Descals, A., Filella, I., & Peñuelas, J. (2022). Nonlinear thermal responses outweigh water limitation in the attenuated effect of climatic warming on photosynthesis in northern ecosystems. *Geophysical Research Letters*, 49, e2022GL100096. https://doi.org/10.1029/2022GL100096

Received 21 JUN 2022 Accepted 7 AUG 2022

Author Contributions:

Conceptualization: Gaofei Yin Funding acquisition: Aleixandre Verger, Josep Peñuelas Investigation: Gaofei Yin Methodology: Gaofei Yin, Josep Peñuelas Supervision: Aleixandre Verger, Josep Peñuelas Writing – original draft: Gaofei Yin Writing – review & editing: Aleixandre Verger, Adrià Descals, Iolanda Filella, Josep Peñuelas

© 2022. American Geophysical Union. All Rights Reserved.

Nonlinear Thermal Responses Outweigh Water Limitation in the Attenuated Effect of Climatic Warming on Photosynthesis in Northern Ecosystems

Gaofei Yin¹, Aleixandre Verger^{2,3}, Adrià Descals^{3,4}, Iolanda Filella^{3,4}, and Josep Peñuelas^{3,4}

¹Faculty of Geosciences and Environmental Engineering, Southwest Jiaotong University, Chengdu, China, ²CIDE-CSIC, València, Spain, ³CREAF, Cerdanyola del Vallès, Barcelona, Spain, ⁴CSIC, Global Ecology Unit CREAF-CSIC-UAB, Barcelona, Spain

Abstract Plant photosynthesis is strongly limited by temperature at high northern latitudes. The temperature sensitivity of plant photosynthesis in scenarios of future climatic warming, however, is highly uncertain. We used sun-induced fluorescence—a satellite proxy for plant photosynthesis—to analyze the spatiotemporal response of photosynthesis to temperature at high northern latitudes. We detected a widespread decline in the sensitivity of photosynthesis to temperature during the last two decades: 3.25 versus 2.19%/°C, in the 2000s and 2010s, respectively. We used methods of machine learning and temporal composition to characterize the contribution of nonlinear thermal responses and water limitation. Both methods consistently identified the nonlinear thermal responses as the main driver of the attenuated positive warming effect. We highlighted the traditionally temperature-limited ecosystems at high northern latitudes may be approaching the temperature tipping point, beyond which the warming effect on plant photosynthesis would transition from positive to negative.

Plain Language Summary Terrestrial plants absorb atmospheric carbon dioxide (CO₂) by photosynthesis, which is a major mechanism mitigating climate change. Plant photosynthesis is strongly limited by temperature at high northern latitudes. Climate warming has decreased the temperature limitation on plant photosynthesis, and induced an increase in photosynthesis. Recent studies revealed a decline in the sensitivity of photosynthesis to temperature with ongoing warming. However, the underlying mechanism is still not clear. We used satellite-observed sun-induced fluorescence data to analyze the spatiotemporal response of photosynthesis to temperature at high northern latitudes. We confirmed the widespread decline in the temperature sensitivity of photosynthesis during the last two decades: 3.25 versus 2.19%/°C, in the 2000s and 2010s, respectively. We further provide compelling empirical evidence that the nonlinear thermal responses (direct effect of warming) outweighed water limitation (indirect effect of warming), accounting for this attenuated effect. We suggested that ecosystems at high northern latitudes are approaching the margin of the benefit of climatic warming.

1. Introduction

Terrestrial plants absorb carbon dioxide (CO_2) from the atmosphere by photosynthesis, which is a major mechanism that offsets anthropogenic carbon emissions and mitigates climate change (Anav et al., 2015; Keeling et al., 1996; Running, 2012). Temperature plays a critical role in controlling plant photosynthetic activity and is therefore one of the main determinants of the global carbon balance (Berry & Bjorkman, 1980; Jung et al., 2017). Anthropogenic climatic warming has decreased the limitation of temperature on plant photosynthesis, especially in high-latitude ecosystems (Huang et al., 2019; Keenan & Riley, 2018), and has exerted positive effects on photosynthesis, indicated by the widely reported global greening (C. Chen et al., 2019; Lucht et al., 2002; Myneni et al., 1997; Piao et al., 2020; Zhu et al., 2016). This warming-induced increase in photosynthesis has further provided a negative feedback to global warming (Zeng et al., 2017).

Anthropogenic warming will continue in the future, but the corresponding response of plant photosynthesis to increasing temperatures is highly uncertain. Several recent studies reported that the greening trends have stalled or even reversed (Berner et al., 2020; Pan et al., 2018; Yuan et al., 2019; A. Z. Zhang et al., 2021). Analyses of the correlation between temperature and photosynthesis, represented by several indicators including satellite-recorded greenness, tree-ring and atmospheric CO_2 measurements, have also indicated weakening

positive or emerging negative relationships (D'Arrigo et al., 2004; Piao et al., 2017; Piao et al., 2014; Wang et al., 2018). These multiple lines of evidence together suggest an attenuated sensitivity of plant photosynthesis to temperature, that is, the magnitude of photosynthetic responses per °C, due to global warming.

Our limited understanding about the effect of warming on plant photosynthesis leads to large uncertainties in comprehending carbon-climate feedbacks and climatic projections (Friedlingstein et al., 2014; Hu et al., 2021). Two main non-exclusive hypotheses may account for the smaller effect of climatic warming on plant photosynthesis: (a) the hypothesis of nonlinear thermal responses, where photosynthetic activity responds parabolically to temperature, that is, photosynthesis first increases with temperature when the ambient temperature is below the optimal temperature, and this increase decelerates and then reverses when the ambient temperature approaches and surpasses the optimum temperature, because electron transport and Rubisco enzymatic capacities become impaired (Berry & Bjorkman, 1980; Huang et al., 2019; Yamori et al., 2014), and (b) the hypothesis of water limitation, where climate warming will incur an exponential increase in the vapor-pressure deficit (VPD) and then cause water stress, because the increased VPD generally causes more atmospheric demand than input from precipitation (Grossiord et al., 2020; Sherwood & Fu, 2014). The photosynthetic rate would decrease under drought conditions due to stomatal closure to prevent hydraulic failure (McDowell et al., 2011).

Elucidating the roles of these two main hypotheses in the attenuated effect of warming has important implications for the accurate projection of carbon uptake. If water limitation dominated the observed attenuated effect of warming, high northern ecosystems may still benefit the most from future warming, because the effect of warming-induced dryness could be partially mitigated by an increased water-use efficiency if the effect of CO_2 fertilization was sustainable (De Kauwe et al., 2021; Smith et al., 2020). In contrast, if the hypothesis of nonlinear thermal responses prevailed, the attenuation of the effect of a positive photosynthetic response would accelerate, and regions exhibiting negative responses to warming may expand substantially, because the thermal adaptation of plant photosynthesis is much slower than the ongoing rate of warming (A. P. Chen et al., 2021; Duffy et al., 2021).

We aimed to characterize the contribution of nonlinear thermal responses and water limitation to the attenuated effect of warming on photosynthesis in high northern ecosystems. We used the clear-sky sun-induced fluorescence (SIF) of the contiguous SIF (CSIF) data set as a proxy of plant photosynthesis, which was strongly correlated with ecosystem gross primary productivity (GPP) (Y. Zhang et al., 2018). The aridity index, calculated as the ratio of precipitation to potential evapotranspiration both from the ERA5 data set (Hersbach et al., 2020), was used to represent water limitation. The aridity index characterizes the dryness in terms of the ecological water balance, accounting for both water supply and demand. The apparent temperature sensitivity (St), expressed as the slope of the linear regression between SIF and air temperature, was used to represent the effect of warming on plant photosynthesis.

2. Materials and Methods

2.1. Sun-Induced Chlorophyll Fluorescence Dataset

SIF is an electromagnetic signal emitted by chlorophyll molecules after the absorption of solar radiation (Porcar-Castell et al., 2014). SIF has been widely used as a proxy of GPP due to its physiological link to plant photosynthesis (Zarco-Tejada et al., 2013). We used SIF rather than existing satellite GPP products (e.g., MODIS, FluxCom) because this data driven approach avoids the uncertainties associated with the parametrization and calibration of more elaborated GPP models (Y. Zhang et al., 2018). We used clear-sky contiguous SIF (CSIF), with 4-day temporal and 0.05° spatial resolutions, to represent plant photosynthetic activity. CSIF data were generated by a machine-learning method using MCD43C1 C6 reflectance as inputs (Y. Zhang et al., 2018). The machine-learning algorithms were trained over the OCO-2 SIF observations with colocation nadir bi-directional reflectance distribution function adjusted reflectance. We selected clear-sky, rather than all-sky, CSIF since it has a stronger correlation with the original OCO-2 SIF retrievals, and also a stronger correlation with GPP estimates from eddy covariance flux towers (Y. Zhang et al., 2018).

2.2. ERA5 Dataset

We analyzed the $0.1^{\circ} \times 0.1^{\circ}$ ERA5 data set to understand the contributions of nonlinear thermal responses and water limitation to the lower temperature sensitivity. ERA5 is the fifth generation of reanalysis produced by the European Center for Medium-Range Weather Forecasts (Hersbach et al., 2020). We extracted daily meteorological data from the original products with hourly resolutions, with a maximized method for air temperature 2 m above the surface and an accumulated method for precipitation and potential evaporation. The aridity index was calculated as the ratio of precipitation to potential evaportanspiration, both accumulated during the growing season. The index defines water limitation in terms of water balance. High and low values represent wet and arid scenarios, respectively.

We selected aridity index because indicators defined in terms of water balance are suitable to quantify the effects of water limitation on GPP variations (Gampe et al., 2021; Jiao et al., 2021). Several recent papers highlighted the importance of VPD on land carbon dynamics (Grossiord et al., 2020; Lu et al., 2022; Yuan et al., 2019). However, Liu et al. (2020) observed VPD plays a secondary role, than soil moisture, in restraining plant growth. This topic is still highly debated and out of the scope of this study.

2.3. MCD12C Land-Cover Dataset

We focused on natural vegetation to eliminate the influence of human management. The 0.05° MODIS land-cover Collection 6 (MCD12C) with International Geosphere Biosphere Programmed classes (Friedl et al., 2010) was used to exclude croplands from the analysis (see Figure S1 in Supporting Information S1).

2.4. Data Analysis

Our aim was to examine the changes in the sensitivity of plant photosynthesis to temperature across ecosystems at high northern latitudes (> 50° N). The 20-year study period, from 2000 to 2019, represents the maximum temporal overlaps of the CSIF and ERA5 data sets. We harmonized all data sets to a 0.1° latitude-longitude resolution, following the ERA5 grid, to ensure consistency among them. Average values were used for CSIF, and the dominant class was used for MCD12C.

The sensitivity of photosynthesis to temperature was calculated over the growing season. We obtained the growing season for each pixel by first smoothing the original CSIF time series using Savitzky-Golay filtering (J. Chen et al., 2004) and then linearly interpolating the series to daily frequency. The smoothed daily time series was then linearly normalized to [0, 1], and the start and end of the growing season (SOS and EOS) were defined as the dates passing through the dynamic threshold of 20% of the annual amplitude. The yearly SOS and EOS were averaged over 2000–2019 to provide the climatological values for these phenological metrics (see Figure S2 in Supporting Information S1). The apparent temperature sensitivity (St) of photosynthesis was defined as the slope of the linear regression between SIF and temperature, both aggregated over the climatological growing season. For a better comparison, St was normalized by the multi-year averaged growing season SIF. St therefore indicated the change in SIF (in %) per degree warming.

A machine-learning method was developed to identify the contributions of nonlinear thermal responses and water limitation to the lower St. Specifically, a random forest (RF) model with 500 decision trees was trained to learn the transfer relationship from temperature and the aridity index to St. The training data set was composed of St calculated over 2000–2019, and temperature and the aridity index were both averaged over 2000–2019. We applied this trained RF model to the two decades in our study (2000–2009 vs. 2010–2019, referred as St²⁰⁰⁰⁻²⁰⁰⁹ and St²⁰¹⁰⁻²⁰¹⁹, respectively) with corresponding decadal averaged temperature and aridity index as inputs. The results indicated that the trained RF model could successfully reproduce the observed St (see Figure S3 in Supporting Information S1). Two additional versions of St for 2010–2019 were predicted, one with an actual aridity index during 2010–2019, controlling temperature in the case of 2000–2009 (St_{A1}²⁰¹⁰⁻²⁰¹⁹), and the other with actual temperature, controlling the aridity index for 2000–2009 (St_T²⁰¹⁰⁻²⁰¹⁹). Finally, we subtracted St_T²⁰¹⁰⁻²⁰¹⁹ and St_{A1}²⁰¹⁰⁻²⁰¹⁹ from St²⁰⁰⁰⁻²⁰⁰⁹ to obtain the contributions of nonlinear thermal responses and water limitation, respectively, to the inter-decadal decrease in St.

A temporally composite analysis was also used for empirical attribution. For characterizing the influence of nonlinear thermal responses of plant photosynthesis, we sorted the years during 2000–2019 based on the 2-m air



Figure 1. Distribution of the temperature sensitivity of photosynthesis (St). St was calculated as the slope of the linear regression between sun-induced fluorescence (SIF) and air temperature (both averaged over the growing season) during 2000–2019. (a) Spatial distribution of St at high northern latitudes (>50°N). (b) St in the temperature-aridity space. Each climatic bin is defined by 1°C intervals of temperature and 0.1 intervals of the aridity index. The temperature and aridity index were both averaged over the growing seasons from 2000 to 2019 to represent the background climatic conditions.

temperature for each grid and identified the 10 years with the highest temperatures as warm years and the other 10 years as cold years. We then separately calculated St for the warm and cold years. The difference between them was used to characterize the variation in St induced by nonlinear thermal responses. We also used a similar procedure to characterize the influence of water limitation, with the aridity index used to identify wet and dry years.

The optimum temperature was also extracted to compare it with ambient temperature. CSIF time series throughout the entire monitoring period and the corresponding temperature were grouped into 1°C temperature bins for each pixel. The 90% quantile of the CSIF data was used as the response of SIF within each temperature bin, and the running means of every three temperature bins were then calculated to develop the temperature response curve of SIF (Huang et al., 2019). The optimum temperature was identified from the response curve at which SIF was maximized. Details regarding the optimum temperature calculation can be found in Huang et al. (2019). The spatial distributions of optimum temperature and the difference relative to the growing season averaged temperature for 2000–2009 and 2010–2019 were shown in Figure S4 in Supporting Information S1.

3. Results

St during 2000–2019 averaged $3.20 \pm 5.61\%$ /°C (mean $\pm \sigma$) across the study area (>50°N). We observed a latitudinal gradient, with a negative response mostly along the southern boundary of the study area: western and central Europe, and western Siberia (Figure 1a). St was highest mainly in the Arctic, where the growing-season temperature was lowest (see Figures S5a and S5b in Supporting Information S1). The cut-off latitude between the negative and positive responses was about 53°N (Figure 2d). Putting St in a temperature-aridity space identified a gradient of St along the temperature variation across all values of the aridity index (Figure 1b). The impact of the aridity index on St was weaker than the impact of temperature, but aridity index suggested that the direct effect of warming (nonlinear thermal responses) outweighed the indirect effect (water limitation) in the attenuated response of photosynthesis in northern ecosystems to climatic warming.

The spatial distributions of St were very similar during the last two decades (see Figures 2a and 2b), but their magnitudes differed significantly: $3.25 \pm 6.41\%$ /°C for 2000–2009 versus $2.19 \pm 6.20\%$ /°C for 2010–2019 (*t*-test, P < 0.0001). We mapped their difference and observed a widespread decrease in St in the last decade (Figure 2c). The most notable decreases were in eastern Europe and eastern Siberia, where they were accompanied by concordant increases in both temperature and aridity (see Figures S5c and S5f in Supporting Information S1). Note the net ecosystem productivity is also controlled by respiration which increases exponentially with warming. Therefore, the attenuation of the positive effects of warming on net carbon uptake may be greater than revealed here.



Geophysical Research Letters



Figure 2. Decreased temperature sensitivity (St) of photosynthesis at high northern latitudes (> $50^{\circ}N$) during the last two decades (from 2000–2009 to 2010–2019). St was calculated as the slope of the linear regression between sun-induced fluorescence (SIF) and temperature (both averaged over the growing season). Spatial distributions of St for (a) 2000–2009, (b) 2010–2019, and (c) their difference. (d) Average St values by latitude during 2000–2009, 2010–2019, and 2000–2019.

Interestingly, we detected many areas, mainly along the southern boundary of our study area, where St reversed its direction from positive to negative, for example, eastern Europe, western and southern Siberia, and central Canada. The reversal of the direction of response caused a 3° northward shift of the areas benefiting from warming (with positive responses to warming), from 52°N to 55°N (Figure 2d).

We calculated the contributions of nonlinear thermal responses and water limitation to the decreased St using a machine-learning method. The widespread warming and drying trends (see Figures S5c and S5f in Supporting Information S1) make these two factors both negative drivers of the inter-decadal variation in St (Figure 3). The contribution from nonlinear thermal responses was an order of magnitude higher than that from water limitation (-1.05% vs. -0.10%) throughout the high northern latitudes. The influence of water limitation on the decrease in St was generally in grasslands in western Siberia and northeastern Canada, which are sensitive to disturbances in water availability because of their shallow root systems, and exhibited a drying trend during the last decade (see Figure S5f in Supporting Information S1). In comparison, nonlinear thermal responses prominently reduced St, especially in the southern (50°N) and northern (the Arctic regions) boundaries in our study area: the ecosystems near 50°N are close to their optimum temperatures (Liu, 2020), so climatic warming would definitely reduce St regardless of the specific rate (see Figure S4 in Supporting Information S1). The effect of Arctic amplification substantially increased the ambient temperature of plant photosynthesis (see Figure S5c in Supporting Information S1), although it is generally regarded as far below the optimum (Liu, 2020).

The differences in composite decadal St (see Section 2) over contrasting dryness $(2.87 \pm 6.36\%)^{\circ}$ C for dry years vs. $4.36 \pm 5.51\%$ /°C for wet years, Figures 4a and 4b) and warmth $(1.56 \pm 9.29\%)^{\circ}$ C for warm years vs.





Figure 3. Inter-decadal difference (2010–2019 vs. 2000–2009) in temperature sensitivity (St) caused by direct (nonlinear thermal responses) and indirect (water limitation) effects of warming. Inter-decadal differences in St from (a) nonlinear thermal responses and (b) water limitation. (c) Latitudinal distributions of inter-decadal differences in St and those caused by nonlinear thermal responses and water limitation. The contributions of the two effects were identified using a machine-learning method.

 $3.92 \pm 8.11\%$ /°C for cold years, Figures 4d and 4e) were $-1.49 \pm 6.70\%$ /°C and $-2.37 \pm 9.29\%$ /°C, respectively, supporting the stronger contribution of nonlinear thermal responses than water limitation to the decrease in St. The latitudinal distributions of St (See Figures 4c and 4f) indicated that productivity would increase at all latitudes in our study area due to climatic warming if the ambient environment was wet and/or cold. In contrast, the southern boundary of the areas benefiting from warming would retreat to 54°N and 56°N in dry and warm years, respectively.



Figure 4. Temperature sensitivity (St) of photosynthesis in contrasting scenarios of aridity and warmth. Composite patterns of St for the 10 (a) driest and (b) wettest years and (c) their values averaged by latitude. Composite patterns of St for the 10 (d) warmest and (e) coldest years and (f) their values averaged by latitude.

4. Conclusions

Ecosystems at high northern latitudes are limited by temperature, and climatic warming favors vegetation productivity (Keenan & Riley, 2018). We used satellite-observed SIF data to detect the effect of a widespread decrease in temperature sensitivity on photosynthesis in northern ecosystems during the last two decades. We further provide compelling empirical evidence that the nonlinear thermal responses (direct effect of warming) outweighed water limitation (indirect effect of warming), accounting for this attenuated effect. Our study highlights the necessity of current land-surface models to accurately reproduce the sensitivity of photosynthesis to temperature. The thermal adoption of plant photosynthesis at the ecosystem scale is slow and may not even occur (A. P. Chen et al., 2021; Duffy et al., 2021; Huang et al., 2019), so ecosystems at high northern latitudes are approaching the margin of the benefit of climatic warming. Many other factors, for example, nutrient limitation, wild fire, and climatic extremes (Angert et al., 2005; Goetz et al., 2005; Penuelas et al., 2017), may further accelerate the reduction of the beneficial role of climatic warming on plant photosynthesis. Therefore, the temperature tipping point at which future warming would exert a negative role on plant photosynthesis may come earlier than previously expected.

Data Availability Statement

The CSIF data are available at https://osf.io/8xqy6/, the ERA5 data are available at https://cds.climate.copernicus. eu/cdsapp%23%21/dataset/reanalysis%2Dera5%2Dland%3Ftab%3Dform, and the MCD12C data are available at https://lpdaac.usgs.gov/products/mcd12c1v006/.

References

- Anav, A., Friedlingstein, P., Beer, C., Ciais, P., Harper, A., Jones, C., et al. (2015). Spatiotemporal patterns of terrestrial gross primary production: A review. *Reviews of Geophysics*, 53(3), 785–818. https://doi.org/10.1002/2015rg000483
- Angert, A., Biraud, S., Bonfils, C., Henning, C. C., Buermann, W., Pinzon, J., et al. (2005). Drier summers cancel out the CO₂ uptake enhancement induced by warmer springs. *Proceedings of the National Academy of Sciences of the United States of America*, 102(31), 10823–10827. https://doi.org/10.1073/pnas.0501647102
- Berner, L. T., Massey, R., Jantz, P., Forbes, B. C., Macias-Fauria, M., Myers-Smith, I., et al. (2020). Summer warming explains widespread but not uniform greening in the Arctic tundra biome. *Nature Communications*, 11(1), 4621. https://doi.org/10.1038/s41467-020-18479-5
- Berry, J., & Bjorkman, O. (1980). Photosynthetic response and adaptation to temperature in higher plants. Annual Review of Plant Physiology, 31(1), 491–543. https://doi.org/10.1146/annurev.pp.31.060180.002423
- Chen, A. P., Huang, L., Liu, Q., & Piao, S. L. (2021). Optimal temperature of vegetation productivity and its linkage with climate and elevation on the Tibetan Plateau. *Global Change Biology*, 27(9), 1942–1951. https://doi.org/10.1111/gcb.15542
- Chen, C., Park, T., Wang, X., Piao, S., Xu, B., Chaturvedi, R. K., et al. (2019). China and India lead in greening of the world through land-use management. *Nature Sustainability*, 2(2), 122–129. https://doi.org/10.1038/s41893-019-0220-7
- Chen, J., Jonsson, P., Tamura, M., Gu, Z. H., Matsushita, B., & Eklundh, L. (2004). A simple method for reconstructing a high-quality NDVI time-series data set based on the Savitzky-Golay filter. *Remote Sensing of Environment*, 91(3–4), 332–344. https://doi.org/10.1016/s0034-4257(04)00080-x
- D'Arrigo, R. D., Kaufmann, R. K., Davi, N., Jacoby, G. C., Laskowski, C., Myneni, R. B., & Cherubini, P. (2004). Thresholds for warming-induced growth decline at elevational tree line in the Yukon Territory, Canada. *Global Biogeochemical Cycles*, 18(3). https://doi.org/10.1029/ 2004gb002249
- De Kauwe, M. G., Medlyn, B. E., & Tissue, D. T. (2021). To what extent can rising [CO₂] ameliorate plant drought stress? *New Phytologist*, 231(6), 2118–2124. https://doi.org/10.1111/nph.17540
- Duffy, K. A., Schwalm, C. R., Arcus, V. L., Koch, G. W., Liang, L. Y. L., & Schipper, L. A. (2021). How close are we to the temperature tipping point of the terrestrial biosphere? *Science Advances*, 7(3). https://doi.org/10.1126/sciadv.aay1052
- Friedl, M. A., Sulla-Menashe, D., Tan, B., Schneider, A., Ramankutty, N., Sibley, A., & Huang, X. M. (2010). MODIS Collection 5 global land cover: Algorithm refinements and characterization of new datasets. *Remote Sensing of Environment*, 114(1), 168–182. https://doi. org/10.1016/j.rse.2009.08.016
- Friedlingstein, P., Meinshausen, M., Arora, V. K., Jones, C. D., Anav, A., Liddicoat, S. K., & Knutti, R. (2014). Uncertainties in CMIP5 climate projections due to carbon cycle feedbacks. *Journal of Climate*, 27(2), 511–526. https://doi.org/10.1175/jcli-d-12-00579.1
- Gampe, D., Zscheischler, J., Reichstein, M., O'Sullivan, M., Smith, W. K., Sitch, S., & Buermann, W. (2021). Increasing impact of warm droughts on northern ecosystem productivity over recent decades. *Nature Climate Change*, 11(9), 772. https://doi.org/10.1038/s41558-021-01112-8
- Goetz, S. J., Bunn, A. G., Fiske, G. J., & Houghton, R. A. (2005). Satellite-observed photosynthetic trends across boreal North America associated with climate and fire disturbance. *Proceedings of the National Academy of Sciences of the United States of America*, 102(38), 13521–13525. https://doi.org/10.1073/pnas.0506179102
- Grossiord, C., Buckley, T. N., Cernusak, L. A., Novick, K. A., Poulter, B., Siegwolf, R. T. W., et al. (2020). Plant responses to rising vapor pressure deficit. *New Phytologist*, 226(6), 1550–1566. https://doi.org/10.1111/nph.16485
- Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horanyi, A., Munoz-Sabater, J., et al. (2020). The ERA5 global reanalysis. *Quarterly Journal* of the Royal Meteorological Society, 146(730), 1999–2049. https://doi.org/10.1002/qj.3803
- Hu, L., Montzka, S. A., Kaushik, A., Andrews, A. E., Sweeney, C., Miller, J., et al. (2021). COS-derived GPP relationships with temperature and light help explain high-latitude atmospheric CO₂ seasonal cycle amplification. *Proceedings of the National Academy of Sciences of the United States of America*, 118(33). https://doi.org/10.1073/pnas.2103423118
- Huang, M. T., Piao, S., Ciais, P., Penuelas, J., Wang, X., Keenan, T. F., et al. (2019). Air temperature optima of vegetation productivity across global biomes. *Nature Ecology & Evolution*, 3(5), 772–779. https://doi.org/10.1038/s41559-019-0838-x

Acknowledgments

This research was financially supported by the Sichuan Science and Technology Program 2021JDJQ0007 and 2020JDTD0003, the National Natural Science Foundation of China 41971282, the Spanish Government project PID2019-110521GB-100, the Catalan government projects SGR2017-1005 and 2020PANDE00117, and the Fundación Ramón Areces project ELEMENTAL-CLIMATE. This work represents a contribution to the CSIC-PTI TELEDETECT.

- Jiao, W. Z., Wang, L. X., Smith, W. K., Chang, Q., Wang, H. L., & D'Odorico, P. (2021). Observed increasing water constraint on vegetation growth over the last three decades. *Nature Communications*, 12(1), 3777. https://doi.org/10.1038/s41467-021-24016-9
- Jung, M., Reichstein, M., Schwalm, C. R., Huntingford, C., Sitch, S., Ahlstrom, A., et al. (2017). Compensatory water effects link yearly global land CO, sink changes to temperature. *Nature*, 541(7638), 516–520. https://doi.org/10.1038/nature20780
- Keeling, C. D., Chin, J. F. S., & Whorf, T. P. (1996). Increased activity of northern vegetation inferred from atmospheric CO₂ measurements. *Nature*, 382(6587), 146–149. https://doi.org/10.1038/382146a0
- Keenan, T. F., & Riley, W. J. (2018). Greening of the land surface in the world's cold regions consistent with recent warming. *Nature Climate Change*, 8(9), 825. https://doi.org/10.1038/s41558-018-0258-y
- Liu, L., Gudmundsson, L., Hauser, M., Qin, D., Li, S., & Seneviratne, S. I. (2020). Soil moisture dominates dryness stress on ecosystem production globally. *Nature Communications*, 11(1), 4892.
- Liu, Y. (2020). Optimum temperature for photosynthesis: From leaf-to ecosystem-scale. Science Bulletin, 65(8), 601–604. https://doi.org/10.1016/j.scib.2020.01.006
- Lu, H., Qin, Z., Lin, S., Chen, X., Chen, B., He, B., et al. (2022). Large influence of atmospheric vapor pressure deficit on ecosystem production efficiency. *Nature Communications*, 13(1), 1653. https://doi.org/10.1038/s41467-022-29009-w
- Lucht, W., Prentice, I. C., Myneni, R. B., Sitch, S., Friedlingstein, P., Cramer, W., et al. (2002). Climatic control of the high-latitude vegetation greening trend and Pinatubo effect. Science, 296(5573), 1687–1689. https://doi.org/10.1126/science.1071828
- McDowell, N. G., Beerling, D. J., Breshears, D. D., Fisher, R. A., Raffa, K. F., & Stitt, M. (2011). The interdependence of mechanisms underlying climate-driven vegetation mortality. *Trends in Ecology & Evolution*, 26(10), 523–532. https://doi.org/10.1016/j.tree.2011.06.003
- Myneni, R. B., Keeling, C. D., Tucker, C. J., Asrar, G., & Nemani, R. R. (1997). Increased plant growth in the northern high latitudes from 1981 to 1991. Nature, 386(6626), 698–702. https://doi.org/10.1038/386698a0
- Pan, N. Q., Feng, X. M., Fu, B. J., Wang, S., Ji, F., & Pan, S. F. (2018). Increasing global vegetation browning hidden in overall vegetation greening: Insights from time-varying trends. *Remote Sensing of Environment*, 214, 59–72. https://doi.org/10.1016/j.rse.2018.05.018
- Penuelas, J., Ciais, P., Canadell, J. G., Janssens, I. A., Fernandez-Martinez, M., Carnicer, J., et al. (2017). Shifting from a fertilization-dominated to a warming-dominated period. *Nature Ecology & Evolution*, 1(10), 1438–1445. https://doi.org/10.1038/s41559-017-0274-8
- Piao, S. L., Liu, Z., Wang, T., Peng, S., Ciais, P., Huang, M., et al. (2017). Weakening temperature control on the interannual variations of spring carbon uptake across northern lands. *Nature Climate Change*, 7(5), 359. https://doi.org/10.1038/nclimate3277
- Piao, S. L., Nan, H., Huntingford, C., Ciais, P., Friedlingstein, P., Sitch, S., et al. (2014). Evidence for a weakening relationship between interannual temperature variability and northern vegetation activity. *Nature Communications*, 5(1), 5018. https://doi.org/10.1038/ncomms6018
- Piao, S. L., Wang, X., Park, T., Chen, C., Lian, X., He, Y., et al. (2020). Characteristics, drivers and feedbacks of global greening. Nature Reviews Earth & Environment, 1(1), 14–27. https://doi.org/10.1038/s43017-019-0001-x
- Porcar-Castell, A., Tyystjarvi, E., Atherton, J., van der Tol, C., Flexas, J., Pfundel, E. E., et al. (2014). Linking chlorophyll a fluorescence to photosynthesis for remote sensing applications: Mechanisms and challenges. *Journal of Experimental Botany*, 65(15), 4065–4095. https://doi. org/10.1093/jxb/eru191
- Running, S. W. (2012). A measurable planetary boundary for the biosphere. *Science*, 337(6101), 1458–1459. https://doi.org/10.1126/science. 1227620
- Sherwood, S., & Fu, Q. (2014). A drier future? Science, 343(6172), 737-739. https://doi.org/10.1126/science.1247620
- Smith, M. N., Taylor, T. C., van Haren, J., Rosolem, R., Restrepo-Coupe, N., Adams, J., et al. (2020). Empirical evidence for resilience of tropical forest photosynthesis in a warmer world. *Nature Plants*, 6(10), 1225–1230. https://doi.org/10.1038/s41477-020-00780-2
- Wang, T., Liu, D., Piao, S., Wang, Y., Wang, X., Guo, H., et al. (2018). Emerging negative impact of warming on summer carbon uptake in northern ecosystems. *Nature Communications*, 9(1), 5391. https://doi.org/10.1038/s41467-018-07813-7
- Yamori, W., Hikosaka, K., & Way, D. A. (2014). Temperature response of photosynthesis in C-3, C-4, and CAM plants: Temperature acclimation and temperature adaptation. *Photosynthesis Research*, 119(1–2), 101–117. https://doi.org/10.1007/s11120-013-9874-6
- Yuan, W. P., Zheng, Y., Piao, S., Ciais, P., Lombardozzi, D., Wang, Y., et al. (2019). Increased atmospheric vapor pressure deficit reduces global vegetation growth. *Science Advances*, 5(8). https://doi.org/10.1126/sciadv.aax1396
- Zarco-Tejada, P. J., Morales, A., Testi, L., & Villalobos, F. J. (2013). Spatio-temporal patterns of chlorophyll fluorescence and physiological and structural indices acquired from hyperspectral imagery as compared with carbon fluxes measured with eddy covariance. *Remote Sensing of Environment*, 133, 102–115. https://doi.org/10.1016/j.rse.2013.02.003
- Zeng, Z. Z., Piao, S., Li, L. Z. X., Zhou, L., Ciais, P., Wang, T., et al. (2017). Climate mitigation from vegetation biophysical feedbacks during the past three decades. *Nature Climate Change*, 7(6), 432. https://doi.org/10.1038/nclimate3299
- Zhang, A. Z., Jia, G. S., & Ustin, S. L. (2021). Water availability surpasses warmth in controlling global vegetation trends in recent decade: Revealed by satellite time series. *Environmental Research Letters*, 16(7), 074028. https://doi.org/10.1088/1748-9326/ac0b68
- Zhang, Y., Joiner, J., Alemohammad, S. H., Zhou, S., & Gentine, P. (2018). A global spatially contiguous solar-induced fluorescence (CSIF) dataset using neural networks. *Biogeosciences*, 15(19), 5779–5800. https://doi.org/10.5194/bg-15-5779-2018
- Zhu, Z., Piao, S., Myneni, R. B., Huang, M., Zeng, Z., Canadell, J. G., et al. (2016). Greening of the Earth and its drivers. *Nature Climate Change*, 6(8), 791–795. https://doi.org/10.1038/nclimate3004